Author’s response to reviews

Title: Prospective systematic review registration: Perspective from the Guidelines International Network (G-I-N)

Authors:

Philip J Van der Wees (vanderwees@hcp.med.harvard.edu)
Amir Qaseem (agaseem@acponline.org)
Minna Kaila (minna.kaila@kolumbus.fi)
Guenter Ollenschlaeger (ollenschlaeger@azg.de)
Richard Rosenfeld (richrosenfeld@ag.e2.exmx.net)
Board of Trustees of the Guidelines International Network (G-I-N) (office@g-i-n.net)

Version: 4 Date: 10 January 2012

Author’s response to reviews:

Dear madam/sir,

The affiliations of the authors have been changed according to the instructions (department, institute, city, country).

We would like to keep the core group of five authors as primary authors of the manuscript, while writing the paper on behalf of the G-I-N Board of Trustees. I have included the Board of Trustees of the Guidelines International Network as 'sixth author' in the submission system and added an acknowledgement section for the group members including affiliation details. I hope this will be adequate for the publication.

Best wishes

Philip Van der Wees

phone: +1 617 432 3423

cell: +1 857 264 8997

email: vanderwees@hcp.med.harvard.edu